Author's response to reviews

Title: The use of CAM by women suffering from nausea and vomiting of pregnancy

Authors:

Taras Hollyer (thollyer@ccnm.edu)
Heather Boon (heather.boon@utoronto.ca)
Alexia Georgousis (ageorgousis@ccnm.edu)
Michael J Smith (michael_j_smith-@hc-sc.gc.ca)
Adrienne Einarson (einarson@sickkids.on.ca)

Version: 2 Date: 15 May 2002

Reviewer #1 A table has been added (table 5), describing severity of NVP and previous CAM use. The discretionary revision was not attainable as we did not ask for this information.

Reviewer #2, The typos have been corrected.
"steadily increasing use has been deleted
Alternative has been deleted.
Actually these data are applicable in the US as well as Canada as women call our line from all over North America, I added that to the text d.